Author's response to reviews

Title: A qualitative analysis of immigrant population health practices in the Girona Healthcare Region

Authors:

Carme Saurina (carme.saurina@udg.edu)
Laura Vall_llosera (laura.vall_llosera@udg.edu)
Marc Saez (marc.saez@udg.edu)

Version: 4 Date: 28 June 2010

Author's response to reviews: see over
Dear Sir:

Please find attached to this cover letter our changes to the format of the manuscript ‘A qualitative analysis of immigrant population health practices in the Girona Healthcare Region’.

Yours sincerely,

[Signature]

Prof. Carme Saurina, PhD
(on behalf of the rest of authors)
Point-by-point description of the changes made

Formatting changes

Major Revisions

Author list: The list of authors in the manuscript should be written exactly as is in the submission system
We have written the list of authors as you suggest
C Saurina, L Vall-llosera, M Saez

Affiliations: Each affiliation should be written in full, in the following format
Department, Institute, (Street), City, (Postal code), Country
We have completed the affiliations
1.- Research Group on Statistics, Applied Economics and Health (GRECS), University of Girona, Campus de Montilivi, Girona 17071, Spain
2.- CIBER of Epidemiology and Public Health (CIBERESP), Biomedical Research Park, Doctor Aiguader 88, Barcelona 8003, Spain

Minor revisions

Email address
We have used de capital letters as you suggest
Carme Saurina CA: carme.saurina@udg.edu
Laura Vall-llosera LA: laura.vall_llosera@udg.edu
Marc Saez MA: marc.saez@udg.edu

Abstract: Please remove keywords
We have removed the keywords

Running footer and Underlining: Please remove
We don’t have running footer and we don’t have underlining

Figures:
We have eliminated the title and the legend on the figures and we have add after the reference list a heading with the name “figure legends”

Typography:
We have checked our manuscript for any typographical errors